Author's response to reviews

Title: Evaluation of the Rapid Scale-up of Collaborative TB/HIV Activities in TB Facilities in Rwanda, 2005-2009

Authors:

Eric S Pevzner (ecp9@cdc.gov)
Greet Vandebriel (gv2124@columbia.edu)
David W Lowrance (LowranceD@rw.cdc.gov)
Michel Gasana (michelgasana@yahoo.fr)
Alyssa Finlay (avf0@cdc.gov)

Version: 7 Date: 22 June 2011

Author's response to reviews:

Below is how we addressed the issues raised by the editorial staff:

1. author list - fixed the author list so that the online format was same as the manuscript format

2. Structure - fixed the paper structure (e.g., added conclusions section)

3. Authors contributions - added the statement that all authors have read and approved the final version.

4. References - added ref 26 where it was supposed to be included

5. Ref - fixed the link for the UNAIDS document

6. Ref - replaced all et. al. with the authors names

7. Figures - cropped the figures as best I could (graphics are not my expertise)

8. Typography - made minor typographical revisions